Institut for Matematik og Datalogi Syddansk Universitet

April 19, 2012 JFB DM830 Network Security – F12 – Lecture 4

Lecture, April 18

We finished chapter 3, except for the Diffie-Hellman key exchange.

Lecture, April 23

We will finish chapter 3 and cover chapter 4.

Lecture, April 27

We will cover Appendix D in the textbook (see the textbook’s Web page) and begin on chapter 5.

Discussion section: April 30

There will be a presentation on viruses from chapter 10 in the textbook.

1. Do problem 4.1 in the textbook. 2. Do problem 4.2 in the textbook. 3. Do problem 4.5 in the textbook. 4. Do problem 4.6 in the textbook. 5. Do problem 4.8 in the textbook. 6. Do problem 4.10 in the testbook.

7. Find certificates (in viewable form) in your browser. Use a screen dump to print one out. What version of SSL/TLS is used? (Note that in Firefox, you can start by looking through Edit -¿ Preferences -¿ Advanced...)